Author's response to reviews

Title: Type II diabetes and impaired glucose tolerance due to severe hyperinsulinism in patients with 1p36 deletion syndrome and Prader-Willi like phenotype

Authors:

Stefano Stagi (stefano.stagi@yahoo.it)
Elisabetta Lapi (e.lapi@meyer.it)
Marilena Pantaleo (m.pantaleo@meyer.it)
Francesco Chiarelli (chiarelli@unich.it)
Salvatore Seminara (seminara@unifi.it)
Maurizio de Martino (maurizio.demartino@unifi.it)

Version: 2 Date: 27 June 2013

Author's response to reviews:

Dear Editor,

we would like to re-submit the enclosed manuscript "Type II diabetes and impaired glucose tolerance due to severe hyperinsulinism in patients with 1p36 deletion syndrome and Prader-Willi like phenotype" for BMC Medical genetics.

We have corrected the text and added Your suggestion. So, we think that the manuscript is suitable for Your Journal.

Consent Statement: Written informed consent was obtained from the parents of the patient for publication of this case report and any accompanying images.

Competing interests: The authors declare that they have no competing interests.

Financial competing interests: The Authors have any financial and non-financial competing interests in relation to this manuscript.

Authors’ contributions:
Stefano Stagi carried out the endocrinological evaluation, conceived of the study and participated in its design
Elisabetta Lapi, carried out the clinical genetic diagnosis
Salvatore Seminara carried out the endocrinological evaluation
Marilena Pantaleo carried out the molecular genetic studies
Francesco Chiarelli participated in the endocrinological evaluation
Maurizio de Martino participated in the endocrinological evaluation and participated in its coordination
All authors read and approved the final manuscript.
I look forward to hearing from you,
Yours sincerely,

Prof. Francesco Chiarelli
Tel.: +39 055 5662913
Fax: +39 055 5662916
E-mail: chiarelli@unich.it